
\documentstyle[12pt]{article} \addtolength{\textheight}{2in} \addtolength{\textwidth}{1in} \addtolength{\leftmargin}{-0.5in}
\def\course{MAT 324}
\def\semester{Fall 2006}

\voffset -.75in
\hoffset -.5in
\newcommand{\vs}{\vspace{.1in}}
\pagestyle{empty}
\begin{document}

\begin{center}
{\bf  PROBLEM SET 7 - The strong law of large numbers}
\end{center}

\begin{enumerate}

  \item prove the SLLN with hints.

    \item apply it to show a.e. real number has binary expansion 
    with equal number of 0's and 1's.







 
\end{enumerate}

\end{document}
